Author's response to reviews

Title: Acute infarct of the corpus callosum presenting as alien hand syndrome: evidence of diffusion weighted imaging and magnetic resonance angiography

Authors:

Jun Liang Yuan (yjl19821225@yahoo.com.cn)
Shuang Kun Wang (shuangkunwang@126.com)
Xiao Juan Guo (xguo1001@126.com)
Wen Li Hu (huwenli@sina.com)

Version: 4 Date: 5 November 2011

Author's response to reviews: see over
Dear Mr Arnold Bongcayao,

Thank you very much for your letter dated 4 November 2011. Our detailed point-by-point responses to the concerns are as follows.

**Major revisions**

1 **Author list**: the list of authors in the manuscript and the list of authors in the submission system should be identical, both in style and order. The preferred style is: First name Initial Last name(e.g. Joe F Bloggs)

[Reply] This has been done.

2 **Abstract**: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Case presentation, conclusions.

[Reply] This has been done.

3 **Minor revisions** (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above).

**Highlighting/tracking**: Please remove from the manuscript.

**Typography**: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

[Reply] These have been done.

We have revised the manuscript in line with the concerns. If you have any questions, please feel free to contact us.

Best regards,

Junliang Yuan